Author’s response to reviews

Title: Genetic analysis of 62 Chinese families with Duchenne muscular dystrophy and strategies of prenatal diagnosis in a single center

Authors:
Zhengfeng Xu (zhengfeng_xu_nj@163.com)
Jingjing Zhang (18913384682@163.com)
Dingyuan Ma (dingyuanma2002@163.com)
Gang Liu (liugang_516@126.com)
Yuguo Wang (wyg.num1@163.com)
An Liu (fyann1213@163.com)
Li Li (njfybjylily@163.com)
Chunyu Luo (lcly0929@126.com)
Ping Hu (njfybjyhuping@163.com)

Version: 3 Date: 23 Oct 2019

Author’s response to reviews:

1. Authors’ contributions

Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship). Currently, the contributions of authors LL, AL and CYL do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0):

Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it

AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);
AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Please note that acquisition of funding, collection of data or general supervision of the research group, alone, does not usually justify authorship. You may acknowledge individuals who assist in the acquisition of funding or of data or supervise the research group in the Acknowledgments section of your manuscript.

If you wish to amend the list of authors, please use the standardised form which you and your co-authors must complete. The authorship change form can be found at the following link: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5
All instructions can be found on the form, please treat the 'current authorship' section as the original authorship. Please return the form within 14 days by email to the Editorial Office with all author signatures (including those newly added/removed).

Response : Thanks. We have described the contributions of the authors in detail.

2. Declarations heading

Please ensure that you include a ‘Declarations’ heading and that it is placed after the Abbreviations section in your manuscript.

Response : Thanks. We have added the "Declarations" after the Abbreviations sections in our manuscript.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

Response : Thanks. We uploaded the clean manuscript.